Author's response to reviews

**Title:** PRISM (Program of Resources, Information and Support for Mothers) Protocol for a community-randomised trial [ISRCTN03464021]

**Authors:**

Java Exception error '80004005'

`java.lang.RuntimeException: java.sql.SQLException: ORA-01017: invalid username/password; logon denied` 

`/commhead.asp, line 357`
Dear Editor

The changes made to this version are summarised below

1. The reference with ‘et al.’ in it now has all the editors listed
2. The Reference list has been pasted after the Acknowledgements and before the Figure legend
3. Table 1 has been included after the Figure legend and the Table legend
4. One email address has been corrected (Jane Gunn)
5. p2 para 3 (and following pages): response rate has been changed to response fraction, in line with current usage
6. p3 (and following pages): all numbers smaller than 10 are now consistently in words
7. p3 (and following pages):
   - the format, in terms of upper and lower case, for general practitioner (GP), maternal and child health nurse (MCHN), community development officer (CDO), health visitor (HV), research team, steering committee, has been applied consistently through the article
8. p3 (and following pages): hyphenation for community-randomised, cluster-randomised, time-out and so on has been made consistent, as has the use of single quotation marks instead of double
9. p 4 (and following pages) eg has been changed to e.g.
10. p5 final line, following with GPs and MCHNs in PRISM. (added)
11. p6, Aims line 6: deletion of health
12. p8 (and following pages) consistency of punctuation for bulleted lists, which are preceded by a colon and where each bullet point end with a semi-colon
13. p9, para 3, final line: potentially participating replaced by eligible
14. p9, final para: consistency in number format e.g. 1,500
15. p10, para 2, lines 1-3: new description of the criteria for matching communities
16. p10 (and following pages) correction of the symbols used in the confidence and power statement to: \( \alpha=0.05 \) (2-sided), \( \beta=0.20 \)
17. p 10, para 5, line 4: correction of the phrase births x LGA
18. p 10, para 5, line 8: 820, giving 5,740 births in each arm.
19. p 10, para 5, final line, change 62% to 63%.
20. p 11, para 5, line 1: insert and home-birth practitioners after hospital of birth
21. p13, para 3: redrafting of this paragraph to make it clearer and more like the previous two
22. p 15, para 1, line 7: a postal questionnaire (delete s)
23. p 15 para 2, final sentence: add full stop
24. p 16, 5th bullet point: delete (duplication)
25. p 16, final bullet point (and in the following points): replace Mean with mean
26. p 18, para 2, line 6: replace participants with women
27. List of abbreviations: Victorian Perinatal Data Collection Unit added
28. References
   - 15 Abdalla
   - 18 Title of paper into bold
21 Publication details completed
38 Journal name changed to Health Education and Behavior (corrected), correction to page numbers
39 Correct abbreviation for journal
29 Acknowledgements: minor changes where bodies have funded both research and implementation
30 P 23.I have deleted the list of papers submitted

Yours sincerely
Judith Lumley